

Assessing Current Coastal Subsidence at Continental Scale: Insights From Europe Using the European Ground Motion Service

Rémi Thiéblemont, Gonéri Le Cozannet, Robert Nicholls, Jérémy Rohmer, Guy Wöppelmann, Daniel Raucoules, Marcello de Michele, Alexandra Toimil, Daniel Lincke

▶ To cite this version:

Rémi Thiéblemont, Gonéri Le Cozannet, Robert Nicholls, Jérémy Rohmer, Guy Wöppelmann, et al.. Assessing Current Coastal Subsidence at Continental Scale: Insights From Europe Using the European Ground Motion Service. Earth's Future, 2024, 12 (8), 10.1029/2024EF004523. hal-04676460

HAL Id: hal-04676460 https://brgm.hal.science/hal-04676460

Submitted on 23 Aug 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Earth's Future

RESEARCH ARTICLE

10.1029/2024EF004523

Special Collection:

Regional Sea Level Change and Society

Key Points:

- Continental-scale coastal land motion assessment in Europe is performed based on the European Ground Motion Service released in 2022
- European flood-prone coastal cities subside on average at 1 mm/yr (Fennoscandia excluded)
- The geocentric reference frame used to calibrate continental-scale land motion affects the results for geodetic (non-geophysical) reasons

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

R. Thiéblemont, r.thieblemont@brgm.fr

Citation:

Thiéblemont, R., Le Cozannet, G., Nicholls, R. J., Rohmer, J., Wöppelmann, G., Raucoules, D., et al. (2024). Assessing current coastal subsidence at continental scale: Insights from Europe using the European Ground Motion Service. *Earth's Future*, *12*, e2024EF004523. https://doi. org/10.1029/2024EF004523

Received 6 FEB 2024 Accepted 26 JUN 2024

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Assessing Current Coastal Subsidence at Continental Scale: Insights From Europe Using the European Ground Motion Service

Rémi Thiéblemont¹, Gonéri Le Cozannet¹, Robert J. Nicholls², Jérémy Rohmer¹, Guy Wöppelmann³, Daniel Raucoules¹, Marcello de Michele¹, Alexandra Toimil⁴, and Daniel Lincke⁵

¹Department of Risk and Prevention, BRGM, French Geological Survey, Orléans, France, ²Tyndall Centre for Climate Change Research, University of East Anglia, Norwich, UK, ³LIENSs, Université de La Rochelle-CNRS, La Rochelle, France, ⁴IHCantabria-Instituto de Hidráulica Ambiental de la Universidad de Cantabria, Santander, Cantabria, Spain, ⁵Global Climate Forum, Berlin, Germany

Abstract Beside climate-change-induced sea-level rise (SLR), land subsidence can strongly amplify coastal risk in flood-prone areas. Mapping and quantifying contemporary vertical land motion (VLM) at continental scales has long been a challenge due to the absence of gridded observational products covering these large domains. Here, we fill this gap by using the new European Ground Motion Service (EGMS) to assess the current state of coastal VLM in Europe. First, we compare the InSAR-based EGMS Ortho (Level 3) with nearby global navigation satellite systems (GNSS) vertical velocity estimates and show that the geodetic reference frame used to calibrate EGMS strongly influences coastal vertical land velocity estimates at the millimeter per year level and this needs to be considered with caution. After adjusting the EGMS vertical velocity estimates to a more updated and accurate International Terrestrial Reference Frame (ITRF2014), we performed an assessment of VLM in European low elevation coastal flood plains (CFPs). We find that nearly half of the European CFP area is, on average, subsiding at a rate faster than 1 mm/yr. More importantly, we find that urban areas and populations located in the CFP experience a near -1 mm/yr VLM on average (excluding the uplifting Fennoscandia region). For harbors, the average VLM is even larger and increases to -1.5 mm/yr on average. This demonstrates the widespread importance of continental-scale assessments based on InSAR and GNSS to better identify areas at higher risk from relative SLR due to coastal subsidence.

Plain Language Summary Land subsidence increases the risk of flooding in low-lying coastal zones by amplifying relative sea-level rise (SLR). In this study, we assess for the first time current coastal land subsidence at the scale of Europe using the new Copernicus European Ground Motion Service (EGMS) that was released in 2022. Our results suggest that nearly half of the low-lying coastal areas in Europe are currently subsiding at a rate faster than 1 mm/yr on average. We find that coastal subsidence is higher on average in areas hosting more people, urban centers and critical infrastructure. This raises concerns that coastal subsidence, and therefore relative SLR, tends to be underestimated in Europe and presumably in many other regions around the world. Our study demonstrates that emerging continental-scale land motion services such as EGMS are useful to better characterize the issue and anticipate coastal risks and adaptation accordingly.

1. Introduction

While the understanding and modeling of sea-level rise (SLR) due to ocean density and mass changes have greatly improved over the past few decades, relative SLR contributions due to vertical land motions (VLMs) remain a major source of uncertainty (Fox-Kemper et al., 2021; Slangen et al., 2023). It is critical to downscale global and regional SLR to local relative sea-level change as this is what causes coastal impacts and adaptation needs (Nicholls, Hanson, et al., 2021). In particular, land subsidence amplifies relative SLR, which in turn can strongly exacerbate exposure to coastal flooding, saltwater intrusion, erosion and loss of wetlands, and damage to infrastructure (Cooley et al., 2022; World Bank, 2010).

Nicholls, Lincke, et al. (2021) found that despite an average global relative SLR of 2.6 mm yr⁻¹ over the past two decades, coastal inhabitants are preferentially located in subsiding locations, and experience an average relative SLR up to four times faster (7.8–9.9 mm yr⁻¹). Locally, land subsidence of several meters can be observed over

the 20th/early 21st century in major coastal cities founded on coastal lowlands and deltas around the world, especially in Asia such as Tokyo, Shanghai, Bangkok and Jakarta (Cao et al., 2021; Esteban et al., 2020; Nicholls, 2018), and other major cities such as New Orleans (Burkett et al., 2003) and Vancouver (Samsonov et al., 2014). Recently, Wu et al. (2022) analyzed subsidence rates in 99 coastal cities worldwide and found that for most of them, part of the land is subsiding faster than the global mean sea level is rising due to ocean density and mass change.

Although large cities are closely scrutinized as they concentrate most economic assets, subsidence is also typical across large deltaic regions that concentrate about half a billion people worldwide (Giosan et al., 2014; Minderhoud et al., 2019; Tessler et al., 2015). Erban et al. (2014) found that in the Mekong Delta, subsidence rates exceed the climate-induced SLR rate by an order of magnitude, so by the mid-21st century an additional inundation hazard of ~ 1 m in portions of the Delta is possible. Recently, by combining groundwater extraction scenarios with climate-change driven SLR and a vertically high resolution digital elevation model of the Mekong Delta, Minderhoud et al. (2020) estimated that $\sim 40-70\%$ of the Mekong delta could fall below mean sea level by the end of the 21st century depending on the degree of subsidence rates that are largest in the most populated area of the delta (Krien et al., 2019). Recently, a study covering the entire US Atlantic coast revealed that a subsidence rate exceeding 3 mm/yr affects most coastal areas, including wetlands, forests, agricultural areas, and developed regions (Ohenhen et al., 2023).

The causes of VLM are multiple (Shirzaei et al., 2021). VLM can be driven by natural processes, such as glacial isostatic adjustment (GIA), which manifests itself particularly in the regions that were covered by or adjacent to the thick ice sheets at the last glacial maximum ~21,000 years ago like eastern North American and the Baltic region in Europe (Peltier, 2004; Spada, 2017). Tectonics and earthquakes can drive coastal subsidence at active plate margins (Leeper et al., 2017; Milker et al., 2016), so as sediment natural compaction, which is a dominant factor in major sediment depocenters such as deltas (Meckel et al., 2007; Teatini et al., 2011; Törnqvist et al., 2008; Zoccarato et al., 2018) and a well-known subsidence driver of coastal reclaimed land (e.g., Cavalié et al., 2023; Park & Hong, 2021). Finally, significant subsidence can be triggered by anthropogenic activities such as peat oxidation (e.g., Koster et al., 2018) or extraction of subsurface fluids (e.g., Galloway & Burbey, 2011; Hasan et al., 2023; Loschetter et al., 2015).

In Europe, two large subsiding coastal flood plain (CFP) regions have been long recognized and documented: that is, the North Italian coastal plain and the Netherlands. In both regions, the subsidence is caused by combined natural and anthropogenic effects that act at different timescales. In the Northern Italian coastal plain, a long-term component is controlled by tectonics and geodynamics and sedimentation, active on time spans of about 10⁶–10⁴ years (Carminati et al., 2003; Cuffaro et al., 2010; Kent et al., 2002). In a shorter-term (1,000 to 10,000 years), VLM in this region is also influenced by glaciation cycles and GIA processes (Antonioli et al., 2009; Stocchi et al., 2005). The latter results were mainly derived from sedimentological indicators. Over the recent historical period, various techniques based on leveling (e.g., Gatto & Carbognin, 1981), tide gauge analysis (Carbognin et al., 2004), height benchmarking (Zerbini et al., 2017), Synthetic Aperture Radar (SAR) (Tosi et al., 2013) or global navigation satellite systems (GNSS) (Zanchettin et al., 2021), allowed characterizing anthropogenic influence (namely fluids withdrawal) on the sinking of the North Italian Coastal plain, associated with an increasing threat of flooding (Lionello et al., 2021 and reference herein). In the Netherlands, subsidence is the result of combined influence of GIA (e.g., Meijles et al., 2018 and references herein), physical compaction of peat and its oxidation (e.g., Erkens et al., 2016; van Asselen et al., 2009), shrinkage and compression of clay (Fokker et al., 2019; Verberne et al., 2023), and ground fluid extraction related to gas production (Schoonbeek, 1976).

In addition to these two large European subsidence hot-spots, more local-scale ones such as for example, Thessaloniki (Raucoules et al., 2008; Stiros, 2001) have been reported and are associated with a major increase of flooding. Until recently, such coastal subsidence hot-spots could be identified through dedicated local to regional InSAR studies, GNSS and tide gauge measurements or leveling surveys (Poitevin et al., 2019; Raucoules et al., 2008, 2013; Wöppelmann et al., 2013). However, there has been to date no analysis of coastal subsidence at the whole European coast scale. This is the gap we propose to fill in the present study based on the recently released European Ground Motion Service (EGMS) from the Copernicus program (Costantini et al., 2021).

This study delivers the first consistent picture of the current state of VLM in low-lying coastal zones prone to marine flooding at the European scale. By doing so, we aim to demonstrate the usefulness of continental scale



Tabla	1
1 able	J

Description of Vertical Land Motion Velocities Estimates Used and Calculated in This Study

Notation	Description	Comments
V _{EGMS}	Vertical velocity estimates from EGMS Ortho product	Raw (or native) vertical velocity data retrieved from EGMS data set. Section 2.1.1 details the data set.
V _{GNSS}	Vertical velocity estimates from GNSS stations	Raw (or native) vertical velocity data retrieved from various GNSS solutions. Section 2.1.2 details the various GNSS data sets used.
V _{GIA}	Vertical velocity estimates of land motions from GIA model	Raw (or native) vertical velocity data retrieved from one GIA model. Section 2.1.3 details the GIA model.
V _{EGMS/} ADJ	Vertical velocity estimates from EGMS adjusted to ITRF2014 reference frame	Post-processed EGMS vertical velocity estimates according to the method described in Section 3.1.
V _{residual}	$\rm V_{EGMS/ADJ}$ – $\rm V_{GIA}$ or vertical velocity with GIA contribution removed.	Vertical velocity estimates from EGMS adjusted to ITRF2014 reference frame minus velocity estimates from GIA model.
$\overline{V}\left(\widetilde{V}\right)$	Mean (median) spatial velocity estimates computed over more than 1 pixel.	\overline{V}_{EGMS} indicates for example, the spatial average of raw vertical velocity of all EGMS pixels over a defined area (e.g. one coastal flood plain).

land motion assessments to better characterize areas at risk from subsidence in coastal zones exposed to SLR and identify challenges. To do so, we use the VLM data set provided by the InSAR-based EGMS service over the 2015–2021 period (Costantini et al., 2021). While our analysis allows the identification and analysis of well-known coastal subsidence hot-spots, it also allows the identification of unrecognized subsiding local scale coastal zones for which flood risk may become a greater concern than expected due to climate change alone. Ultimately, this study supports the development of climate services for coastal adaptation to SLR by providing a first assessment of where subsidence needs to be considered and added to sea-level scenarios in Europe.

The paper is organized as follows. First we describe the data sets used in this study to characterize VLM, coastal zones and associated economic and natural assets. Then, we make a comparison of EGMS VLM estimates with independent GNSS estimates and describe how VLM in CFPs are derived. The results are then presented at various spatial scales in units of CFP but also with respect to exposed assets within these CFPs. We conclude by an extended discussion on the implications of this work for future analyses in Europe and other regions of the world.

2. Data

In this study, we characterize contemporary VLM velocities within the European CFP using EGMS. EGMS InSAR-based estimates are further compared with GNSS products. Exposed assets and CFP data sets are also described below.

2.1. Vertical Land Motion

Three different types of data are used to characterize and quantify VLM: InSAR-based EGMS, GNSS-based velocity estimates and a GIA ensemble model. The terminology of the various vertical land velocity estimates employed hereinafter in this paper is summarized in Table 1.

2.1.1. EGMS

EGMS uses Synthetic Aperture Radar Interferometry (InSAR) data derived from Sentinel-1 to detect and measure ground movements across Europe with millimeter precision. The EGMS aims to provide consistent, updated, standardized, harmonized across national borders and reliable information regarding natural and anthropogenic ground motion phenomena over Europe (Crosetto & Solari, 2023). EGMS relies on the advances and improvements of the advanced Differential InSAR (A-DInSAR) data processing and analysis techniques, also referred to as Persistent Scatterer Interferometry (see Crosetto et al., 2016, 2021). EGMS products are being made by the consortium ORIGINAL (OpeRatIonal Ground motion INsar Alliance) comprising four different InSAR Processing Entities using methods proven over the two last decades. All algorithms and methods used to produce EGMS products are detailed in the document *Algorithm Theoretical Basis Document* (Ferretti et al., 2023).

Although the A-DInSAR technique is powerful to measure localized deformation, it can only provide spatially relative measurements on local spatial scales up to tens of km. On the other hand, GNSS measurements can be considered absolute and accurate over large spatial scales, but are very sparse. In this regard, EGMS exploits the synergy between InSAR and GNSS by calibrating the A-DInSAR product with a low-frequency deformation map obtained from a GNSS gridded model (Crosetto et al., 2020).

EGMS delivers three products that are updated annually:

- The Basic product (or Level 2a) which provides line of sight velocity maps in ascending and descending orbits with annotated geolocation and quality measures per measurement point. Basic products are referred to a local reference point.
- The Calibrated product (or Level 2b) also providing line of sight velocity maps in ascending and descending orbits but referenced to a model derived from GNSS time-series data so that the product is absolute in a geocentric sense (i.e., no longer relative to a local reference point but relative to a geocentric reference frame).
- The Ortho product (or Level 3), constructed from the Calibrated product using trigonometric calculation, which provides horizontal and vertical land velocity components resampled on a 100 m resolution grid and anchored to the geocentric reference frame model.

The EGMS calibration from Level 2b is obtained by referencing the Basic product (Level 2a) to the Augmented EUREF Permanent Network (A-EPND) model derived from GNSS data. The A-EPND gridded model is constructed based on GNSS solution data from the EUREF Densification product (EPND) mainly (https://epnd.sgo-penc.hu/) and completed by input GNSS data from Nevada Geodetic Laboratory (NGL). The A-EPND model is calculated on a regular 50-km grid covering the whole Europe (~3,800 grid points) and results in a long-wavelength (>50 km) 3D velocity field in the ETRF2000 realization of the ETRS89 reference frame. The Calibrated product is then obtained from the Basic product by calibrating its lower-frequency phase trends using the A-EPND model. The calibration procedure is based on common existing methodologies (see e.g. Farolfi et al., 2019). More information on the A-EPND model and calibration procedure are provided in the EGMS *Algorithm Theoretical Basis Document* (Ferretti et al., 2023), the EGMS *GNSS Calibration Report* (Larsen et al., 2023) and in the Text S1 in Supporting Information S1.

Our study relies on the first update of the Ortho product (Level 3) covering the period 2015–2021 (released in 2022). We focus on the vertical component. The Ortho data set was obtained from the EGMS web portal (https:// egms.land.copernicus.eu/), which provides 100 km × 100 km geotiff rasters of the vertical velocity best estimate over the period 2015–2021. The EGMS model applied to estimate the temporal coherence of the measured point and for computing the root mean square error of each displacement time series is a third order polynomial, plus a seasonal component (sinusoidal). Each 100 km × 100 km geotiff raster is accompanied with a tabular file (format. csv) providing, for each pixel (100 m × 100 m) of the raster file, the spatial coordinates, the root mean square error of the time series residuals from the fitting procedure, the mean vertical velocity, acceleration and seasonality, and their standard error, respectively, and the full vertical position time series. In this study, we relied on the contemporary mean vertical velocity estimates by the EGMS fitting model at each pixel (hereinafter V_{EGMS}, see Table 1). Overall, the Ortho product provides a mean velocity estimate with a precision better than 1 mm/yr (Capes & Passera, 2023; Sala Calero et al., 2023).

To cover the entire European coastline, we downloaded 415 EGMS tiles of 100 km × 100 km spatial extent that we merged into one large pan-European raster file and clipped to the European coastal mask defined within the Dynamic Interactive Vulnerability Assessment (DIVA) modeling framework (see further details in Section 2.2). This leads to more than $10^6 V_{EGMS}$ (i.e., 10^6 filled pixels) at the scale of Europe.

2.1.2. Global Navigation Satellite System

Contemporary V_{EGMS} are compared with vertical velocity estimates from continuously operating GNSS stations (hereinafter V_{GNSS} , see Table 1) in Europe. We used V_{GNSS} distributed by the SONEL (Système d'Observation du Niveau des Eaux Littorales) web portal (https://www.sonel.org/?lang=en). SONEL serves as the GNSS data assembly center for the Global Sea Level Observing System (GLOSS), which is developed under the auspices of the IOC/UNESCO. This data assembly center, hosted at La Rochelle University in France (Wöppelmann et al., 2021), aims to provide high-quality continuous measurements of sea- and land-levels at the coast from tide

gauges (French-only scope) and modern geodetic techniques, especially permanent GNSS stations co-located at or nearby tide gauges (global scope).

The SONEL portal currently distributes V_{GNSS} at coastal stations worldwide from four different solutions (groups): ULR7A from University of La Rochelle (Gravelle et al., 2023), NGL14 from NGL (Blewitt et al., 2016, 2018), JPL14 from Jet Propulsion Laboratory (Heflin et al., 2020) and GT3 from the German Research Center for Geosciences in Postdam (Männel et al., 2022). The four solutions express their vertical velocity field with respect to the International Terrestrial Reference Frame 2014 (ITRF2014) (Altamimi et al., 2017). In spite of adopting the same International GNSS Service standards and conventions, the different analyst choices (beyond the agreed standards and conventions) at any of the major GNSS processing steps can lead to significant differences in vertical velocity estimates between solutions for a same GNSS station (e.g., Ballu et al., 2019). Here, we compared V_{EGMS} with V_{GNSS} from ULR7A and NGL14 solutions as these solutions offered the largest amount of coastal GNSS stations to compare with, and represent two independent GNSS data analysis strategies, the so-called "double difference" and "precise point positioning" in GNSS jargon.

Finally, our analysis of V_{GNSS} also investigates vertical velocity estimates of the EUREF Permanent GNSS Network (EPN) solution (Bruyninx et al., 2019) as this solution was used for calibrating V_{EGMS} (Larsen et al., 2023). More specifically, we used the EPN D2150 densification solution (Kenyeres et al., 2019), which retrieved vertical velocities for nearly 2,500 stations in Europe.

2.1.3. GIA

Ongoing changes in the solid Earth caused by the viscous adjustment of the mantle to the substantial mass redistribution that followed the Last Glacial Maximum (named Glacial Isostatic Adjustment; GIA) have an important contribution to VLM in Europe (Peltier & Tushingham, 1991). Typically, the GIA causes a land uplift in the Fennoscandia region (contributing to relative sea-level fall) and a subsidence in its near periphery (i.e., a relative SLR); for example, The Netherlands coast (Vermeersen et al., 2018). Despite the well-established effect of the GIA in Northern Europe, there remain uncertainties on its magnitude and spatial distribution that are reflected by the substantial discrepancies between the output of various existing GIA models (e.g., Bradley et al., 2011, 2023; Kuchar et al., 2012; Lambeck et al., 1998; Meijles et al., 2018; Peltier et al., 2015; Simon et al., 2018).

Here, we rely on the GIA model of Caron et al. (2018), which was built from an ensemble of 128,000 GIA simulations. Ensemble simulations are designed by varying solid-Earth parameters (lithosphere thickness and mantle viscosities) and amplitudes of global deglaciation histories over the past 20,000 years. Each GIA ensemble member provides a consistent set of changes in relative sea level, solid-Earth deformation and changes in equivalent water height and comes with a likelihood that reflects how good the fit is to a data set of vertical GNSS velocities and paleo sea-level records. Therefore, the model allows for a robust quantification of the uncertainties associated with GIA. The spread between the ensemble members depicts the uncertainty in the GIA predictions due to uncertainty in the solid-Earth parameters and the deglaciation history. This GIA model was used to reconstruct and successfully close the basin-mean sea-level budget over the 20th century (Frederikse et al., 2020). The gridded GIA data set of corresponding relative sea level and solid earth deformation mean rates and uncertainty are retrieved from Frederikse, Landerer, and Caron (2019).

2.2. Coastal Mask and Flood Plains

Since the study focuses on coastal regions exposed to coastal flooding and SLR, we restrict our analysis to coastal areas and CFPs. The coastal mask and related CFPs were extracted from the DIVA model (Lincke et al., 2022). DIVA has been developed for broad-scale coastal assessment of SLR and used to assess issues such as coastal erosion (Hinkel et al., 2013), coastal flooding (Hinkel et al., 2014), coastal wetland change (Schuerch et al., 2018), coastal migration and adaptation (Lincke & Hinkel, 2021) and subsidence/relative SLR (Nicholls, Lincke, et al., 2021).

More specifically, we used the DIVA coastal mask to construct a V_{EGMS} coastal raster (i.e., restricting EGMS coverage to the European coastal region). The DIVA coastal mask allows identifying coastal area, which means hydrologically connected to the ocean by an elevation below 20 m or within 1 km distance to the coastline. The



Earth's Future



Figure 1. Location of coastal flood plains in Europe as defined in this study.

resulting V_{EGMS} coastal raster covers therefore the entire European coastline (i.e., not solely restricted to low-lying CFPs) and is used for V_{EGMS}/V_{GNSS} comparisons (see Section 3.1).

Now to focus on flood-prone areas, a CFPs vectorized layer developed within the frame of the Coastal Climate Core Services H2020 project (CoCliCo; https://coclicoservices.eu/) is used in addition. CFPs are contoured using the 30-m resolution Copernicus Digital Surface Model GLO-30 (European Space Agency Sinergise, 2021), considering a bathtub flooding approach (including hydraulic connections) calculated for a 1-in-100-year event combined with a 2-m SLR. This results, at the scale of Europe, in a total of about 41,000 CFPs spanning a wide range of sizes with a minimum CFP area of 0.01 km^2 (i.e., equivalent to one EGMS pixel size) and a maximum CFP area of $17.6 \times 10^3 \text{ km}^2$. The median CFP area is 0.06 km^2 while the mean CFP area is 2.5 km^2 : that is, the distribution of CFPs is strongly positively skewed. The European CFP coverage is shown on Figure 1.

2.3. Exposure

The pan-European Corine Land Cover (CLC) for the reference year 2018 (CLC2018) is used to characterize land use and land cover within CFPs. The CLC nomenclature consists of 44 land use and land cover classes (Bossard et al., 2000; Feranec et al., 2016). The classes are defined and explained in an illustrated technical guideline (Kosztra et al., 2017). The CLC2018 raster layer is provided at a minimum feature width of 100 m at the scale of Europe, which has the same resolution as the EGMS Level-3 Ortho product. Note that among the 44 land classes, we focus mainly on infrastructures assets (i.e., urban fabrics, networks, commercial units, airports/harbor) and typical coastal natural environments (e.g., marshes).



Figure 2. (a) Scatter plot of V_{GNSS} from ULR7A solution and corresponding V_{EGMS} estimates. The black dashed line displays the identity line and the red solid line displays the results of the linear regression between V_{EGMS} and V_{GNSS} (the associated equation and correlation coefficient are provided in the legend) (b) Regional example of global navigation satellite systems (GNSS) stations locations (triangles) and V_{EGMS} spatial coverage around the Bay of Biscay (western France and northern Spain Atlantic coast). Ninety-three coastal GNSS stations are available at the scale of Europe for the ULR7A solution. The green contour delimits the Arcachon bay area, which is further analyzed in Figure 4. V_{EGMS} from the scatter plot (a) and pixels on the map (b) have the same color ranges.

To analyze the exposed population, we used the population gridded data from Eurostat for the reference year 2021. The total EU population measured on the 1 km² population grid in the census reference year 2021 was 443.2 million residents. The data sets are based on data from the 2021 population and housing census at the level of 1 km² grid cells (Lutz et al., 2019).

3. Methods and Validation

In this section, we compare vertical land velocities estimates obtained from EGMS (V_{EGMS}) and GNSS stations (V_{GNSS}) and we detail the method that has been developed to calculate VLM in CFP. Hereinafter, subsiding and uplifting land motion velocities are displayed in blue and red, respectively.

3.1. Comparison Between EGMS and GNSS Vertical Velocities

EGMS Level-3 Ortho is a highly relevant and unique product to characterize the spatial distribution of current VLM as its coverage and resolution is unprecedented at the scale of Europe. This contrasts with GNSS station networks that provide precise and often long-term records (some records are longer than 20 years) of vertical land velocity estimates that are however sparsely distributed spatially. SONEL V_{GNSS} estimates are independent from EGMS and thus highly relevant to assess V_{EGMS} estimates.

Figure 2 shows a comparison between V_{GNSS} retrieved from 93 GNSS stations distributed across the European coastline and from the ULR7A solution (available on SONEL), and V_{EGMS} estimates from the EGMS pixels in which the GNSS stations are located. An example of the spatial location of GNSS stations and the EGMS spatial coverage along the Bay of Biscay coastal region is shown on Figure 2b. The scatter plot (Figure 2a) shows a good correlation between V_{GNSS} and V_{EGMS} estimates (Pearson's correlation coefficient of 0.94). Similarly, the correlation coefficient between V_{GNSS} from the NGL14 solution (171 stations, available on SONEL) and V_{EGMS} is 0.92 (Figure S1 in Supporting Information S1). Note that this analysis was repeated by considering the averaged V_{EGMS} over pixels located within a 200 m radius around the GNSS stations and the results of the comparison were very similar (Figure S2 in Supporting Information S1).

) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creat

23284277, 2024, 8, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024EF004523 by BRGM, Wiley Online Library on [23/08/2024]. See the Terms and Co





Figure 3. Analysis of EPN global navigation satellite systems vertical velocity differences in the ITRF2014 and ETRF2000 geodetic reference frames in Europe. (a) European mapping of the vertical land velocity differences. (b) Vertical land velocity difference as a function of the latitude.

Our comparison hence reveals an overall good agreement between V_{EGMS} and V_{GNSS} , however, the linear regression analysis between both estimates suggests that V_{EGMS} tends to be more negative than V_{GNSS} as suggested by the intercept of -0.89 mm/yr, and this effect strengthens for large positive land motion velocities (i.e., large ground uplift region). Similar offsets are found when regressing V_{EGMS} over V_{GNSS} from the NGL14 solution (Figure S1 in Supporting Information S1). This seeming negative V_{EGMS} bias against V_{GNSS} is investigated in more detail below.

As mentioned in 2.1.1 and 2.1.2, V_{EGMS} are calibrated with respect to EUREF GNSS network (EPN) in the European geodetic reference frame ETRF2000. In this reference frame, the Southern Europe appears to experience an average subsidence of between 0.5 and 1 mm/yr (see EGMS GNSS Calibration report; Larsen et al., 2023). Figure 2b reflects well this effect and shows that most EGMS pixels from the Bay of Biscay coastal region have a V_{EGMS} comprised between -2 and 0 mm/yr (orange and yellow pixels).

Next, we compare V_{GNSS} from the EPN D2150 densification solution expressed in the ETRF2000 geodetic reference frame (and used to calibrate V_{EGMS}) to those of the same GNSS stations but expressed with respect to the ITRF2014 (international geodetic reference frame, common to ULR7A and NGL14 solutions available on SONEL). The results of this comparison are shown in Figure 3. It appears that V_{GNSS} from EPN D2150 expressed in the ETRF2000 geodetic reference frame are systematically more negative than when expressed in the ITRF2014, and that this effect amplifies toward higher latitudes. This result is consistent with the previous (see Figure 2a) comparative analysis of V_{GNSS} from ULR7A and NGL14 solutions (expressed in the ITRF2014) and V_{EGMS} (calibrated in the ETRF2000). Also, the strengthening of the deviation toward large land motion uplift (Figure 2a) is consistent with the amplifying differences in high latitudes of Europe where the land motion uplift caused by GIA is large (Figure 3a).

The ITRF is the most precise and stable geocentric reference frame achieved so far by the international community within the International Earth Rotation and Reference Systems framework (see https://www.iers.org/) and is regularly updated (e.g., ITRF2000, ITRF2005, ITRF2008, ITRF2014 and ITRF2020). Note that the 2000 (2005, etc.) in ITRF2000 underlines the latest geodetic observations considered in the ITRF release. Although not the most recent one, the ITRF2014 still remains a reference frame solution in the data analysis strategies of most groups (e.g., ULR, JPL, NGL) for which highly demanding applications in terms of precision and stability of positioning are required. In this regard, we apply a simple transfer function allowing to adjust V_{EGMS} calibrated within ETRF2000 into the ITRF2014 ($V_{EGMS/ADJ}$ hereinafter) to illustrate the importance of the geodetic reference frame choice on EGMS calibration to analyze coastal VLM. The transfer function is shown on Figure 3b, which shows that the difference of vertical velocities estimates between ETRF2000 to ITRF2014 can





Figure 4. Sketch illustrating the method developed to calculate the coastal flood plains (CFPs) average vertical land velocity from European Ground Motion Service (EGMS) data. (a) EGMS pixels and its associated V_{EGMS} (100 m × 100 m resolution), and corresponding spatial coverage of EGMS pixels within each CFP (in percent of CFP area). (b) \overline{V}_{EGMS} , that is, mean V_{EGMS} over all EGMS pixels within each CFP. (c) $\overline{V}_{EGMS/ADJ}$, that is, same as (b) but adjusted to the ITRF2014 reference frame.

be adequately modeled by a second order polynomial with respect to the latitude (the determination coefficient of the fit, or R^2 , is 0.99). Note that for the purpose of this study, such a level of precision in the predictive model (root mean square error lower than 0.02 mm/yr) is sufficient.

3.2. Assessing VLM in CFPs

Figure 4 illustrates the method that we developed to assess CFP VLM based on contemporary V_{EGMS} estimates. The Arcachon bay case is used as an example (see also Figure 2b).

For each CFP, we first calculate the fraction of spatial area that is covered by EGMS pixels. Figure 4a reveals that the pixel coverage can vary substantially from one CFP to another. For instance, the island located in the middle of the bay (called "l'île aux oiseaux"), which is one entire CFP, is covered by EGMS pixels over less than 20% of its area (lighter gray contour). In contrast, the narrow CFPs located in the northern part of the bay are almost entirely covered by EGMS pixels (more than 80%, black contours). The largest CFP of the bay of Arcachon, along the southern shore, has an EGMS pixel coverage ratio of nearly 50%.

The ratio of EGMS pixels depends on the nature of land coverage and its capacity to reflect the radar signal, but also depends on the area of CFP. Typically, urban areas are very well covered by EGMS pixels in contrast to



Table 2 CFPs Coverage by European Ground Motion Service Data (Total; 40,746 CFPs)						
EGMS coverage	#CFPs	Total area (10^3 km^2)	Mean CFP area (km ²)	Median CFP area (km ²)	Max CFP area (km ²)	
]0; 20]%	2,886	69.0	24	0.5	17.5×10^{3}	
]20; 40]%	3,068	20.6	6.7	0.1	6.0×10^{3}	
]40; 60]%	2,999	5.2	1.7	0.1	2.4×10^{3}	
]60; 80]%	3,900	1.2	0.3	<0.1	154	
]80; 100]%	16,613	2.1	0.1	<0.1	50	

agricultural land or forest areas. The ratio of EGMS pixels for all 40,746 European CFPs is reported in Table 2. Nearly one quarter of CFPs are empty of EGMS pixels, which however, corresponds to less than 5% of the total CFP coverage in Europe and concerns mostly Norway, Sweden and Finland. Table 2 also shows that CFPs with the lower EGMS pixel coverage ratio are also the most extended ones; they correspond typically to very large low-lying areas that embed various land uses such as infrastructures, networks and urban areas surrounded by agricultural land areas (e.g., in the Netherlands). Therefore, for extended CFPs with a low EGMS coverage ratio, the V_{EGMS} estimates may not necessarily be representative of the entire CFP behavior, but rather reflect the VLM state of urban assets within those large CFPs.

To obtain an overall view of the VLM velocities in the CFPs, we derive \overline{V}_{EGMS} by averaging all V_{EGMS} within each CFP (Figure 4b). As illustrated on Figure 4b, we find that the largest CFP of the Arcachon bay has a \overline{V}_{EGMS} of -2.2 mm/yr (red), hence suggesting a significant subsidence. After adjusting to the ITRF2014 reference frame ($\overline{V}_{EGMS/ADJ}$) (see Figure 3), the VLM velocities of the later CFP is reduced to -1.6 mm/yr (Figure 4c, orange). Note that, for each CFP, we also assess the median (noted \widetilde{V} hereinafter, see also Table 1), the first quartile and the third quartile velocities systematically (see Figure S3 in Supporting Information S1). For most CFPs containing EGMS pixels (i.e., ~70%), we found that \overline{V}_{EGMS} is lower than \widetilde{V}_{EGMS} ; this means that the distribution of V_{EGMS} within CFPs are generally skewed toward more subsiding values.

4. Results

Our results are presented in this section. We first analyze the overall state of subsidence in Europe. We then assess in more detail a selection of subsiding coastal hot-spots identified in our data set. Finally, we assess how subsidence may enhance coastal risk in Europe over the 21st Century by combining our EGMS-based data set with assets and exposure inferred from the CLC and population.

4.1. State of Vertical Land Motion of European CFPs

The distribution of mean (\overline{V}) and median (\widetilde{V}) CFP's contemporary VLMs for all of Europe's coast is shown on Figure 5. The distribution is expressed either with respect to the number of CFPs (Figures 5a and 5b), or with respect to the associated CFP area (Figures 5c and 5d). To emphasize the importance of the geodetic reference frame on vertical land velocity estimates, we analyze both the EGMS Ortho product (\overline{V}_{EGMS} and \widetilde{V}_{EGMS} , see also Figure 4b) and the EGMS Ortho product adjusted to ITRF2014 ($\overline{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$, see also Figure 4d). To quantify how the GIA contributes to VLM in Europe, we also calculate and analyze $\overline{V}_{residual}$ and $\widetilde{V}_{residual}$; that is, the mean and median EGMS Ortho product adjusted to ITRF2014 to which the GIA effect on solid-Earth vertical deformation has been subtracted. In addition to mean and median estimates, the spatial variability of vertical velocities estimates within each CFP has been analyzed using the first and third quartiles (Figures S3 and S4 in Supporting Information S1). We find that interquartile width for each CFP remains mostly below 1 mm/yr and tends to increase for large CFPs.

CFPs that are uplifting on average (i.e., $\overline{V} > 0.5$ mm/yr) represent a large majority of CFPs in Europe with more than 20,000 CFPs for both \overline{V}_{EGMS} and $\overline{V}_{EGMS/ADJ}$ estimates (Figure 5a). The same conclusion is draw from median velocity estimates (Figure 5b). Figures 5c and 5d reveals however that despite the large number of uplifting CFPs, the corresponding spatial extent remains modest as it comprises less than 10% of the total European CFP area both \overline{V}_{EGMS} and $\overline{V}_{EGMS/ADJ}$ estimates. This is typically related to small CFPs affected by uplifting effect of the GIA in





Figure 5. Distribution of (a,c) mean (\overline{V}) and (b,d) median (\widetilde{V}) vertical velocities in coastal flood plain (CFP) per (a,b) number of CFPs and (c,d) per ratio of CFP area in Europe. The distribution are calculated from European Ground Motion Service (EGMS) Ortho product (\overline{V}_{EGMS} and \widetilde{V}_{EGMS} , in gray), EGMS Ortho product adjusted to ITRF2014 ($\overline{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$, in black), and EGMS Ortho product adjusted to ITRF2014 to which glacial isostatic adjustment contribution is subtracted ($\overline{V}_{residual}$ and $\widetilde{V}_{residual}$, in white). Note that in panels (c,d), for each type of vertical velocity estimate, the sum over the ranges] ∞ ;0.5], [0.5;-0.5] and]-0.5;-\infty[is 100%.

the Northern European countries. The latter result is also consistent with the fact that when the vertical velocity from GIA model is subtracted from $\overline{V}_{EGMS/ADJ}$ (i.e., $\overline{V}_{residual}$ on Figure 5a) or $\widetilde{V}_{EGMS/ADJ}$ (i.e., $\widetilde{V}_{residual}$ on Figure 5b), a large amount of uplifting CFPs is moved toward more stable conditions (i.e., range -0.5 to 0.5 mm/yr).

In contrast to uplifting CFPs, the number of CFPs that are on average subsiding at a rate faster than 1 mm/yr is low: 5,000 CFPs for \overline{V}_{EGMS} and 2,500 CFPs for $\overline{V}_{EGMS/ADJ}$ (Figure 5a). First, we note that this difference between \overline{V}_{EGMS} and $\overline{V}_{EGMS/ADJ}$ is large, which demonstrates that the geodetic reference frame used to express the vertical velocity has a large effect on CFP VLM assessment and needs to be acknowledged. Second, the 2,500 subsiding CFPs (out of ~30,000 CFPs containing EGMS pixels) determined from $\overline{V}_{EGMS/ADJ}$ represents nearly 50% of the total CFP area in Europe (Figure 5c). Overall, our analysis of $\overline{V}_{EGMS/ADJ}$ suggests that 50% (10%) of the European land area located in CFPs experience an average subsidence stronger than 1 mm/yr (2 mm/yr). We also find that 80% of CFPs



Figure 6. The distribution of $\overline{V}_{EGMS/ADJ}$ in the coastal flood plain (CFP) per European coastal country. The red, white and blue areas indicate the proportion per country of CFP area that are, on average, uplifting (>1 mm/yr), stable (between -1 mm/yr and 1 mm/yr) and subsiding (<-1 mm/yr), respectively.

that have an average subsidence stronger than 1 mm/yr are covered by 75% of pixels with $V_{EGMS/ADJ} < -0.5$ mm/yr (Figure S4 in Supporting Information S1). This means that despite some spatial variability, subsiding CFPs are mostly covered by pixels with negative velocity estimates. Finally, Figures 5c and 5d also reveal that, when subtracting the vertical velocity from the GIA model (see $\overline{V}_{residual}$ or $\widetilde{V}_{residual}$), more than 60% of the CFP total area becomes stable (range -0.5 to 0.5 mm/yr). This indicates that GIA is a first order contributor to both low-lying coastal land uplift and subsidence.

Next, we examine the distribution of $\overline{V}_{EGMS/ADJ}$ in CFPs per European coastal country. Results are shown in Figure 6 (see also Figure 1 for visualizing European CFPs). Here, vertical range velocity classes are simplified as we assume that an uplifting (subsiding) CFP corresponds to a $\overline{V}_{EGMS/ADJ}$ larger (lower) than 1 mm/yr (-1 mm/yr) and is considered as stable otherwise (i.e., in the range -1 to 1 mm/yr).

As expected, CFPs of northern European countries (i.e., Norway, Sweden, Finland and Estonia) are mostly uplifting on average as a consequence of the GIA effect. Just southward of these four uplifting countries, Latvia, Lithuania, Denmark, UK and Ireland show rather stable VLMs, and further south, a widespread subsidence (with exceptions such as Belgium). More specifically, In Germany and the Netherlands, a vast majority (near 75% of the total area) of CFPs are experiencing an average subsidence, which is consistent with the combined effect of GIA and more localized subsidence of other origin (see Introduction). Among the countries of southern Europe with expanded CFPs, Italy, Greece and Romania have more than 75% area of their CFPs that are subsiding on average. Finally, in the Western Europe, more than half of CFPs in France and Portugal are subsiding on average.

The analysis of the distribution of $\widetilde{V}_{EGMS/ADJ}$ in CFPs per European coastal country (i.e., median instead of mean) reveals a slightly different picture (see Figure S5 in Supporting Information S1). While the ratio of uplifting and subsiding CFPs remains nearly unchanged for Fennoscandia and Mediterranean countries, it drastically changes for the Netherlands, Germany and France, where most CFPs now fall into the stable category. This illustrates that the VLM distribution is skewed, which becomes particularly pronounced when CFPs are large (see Discussion section).

4.2. Subsiding Hot-Spots Detected in EGMS

In the previous section, EGMS vertical velocities were processed per CFP to provide an overview of the current state of VLM in low-lying vulnerable coastal regions in Europe. Here, we focus on contemporary V_{EGMS} at a finer scale and illustrate how pixel-scale vertical velocities can characterize subsiding hot-spot regions. Note that $V_{EGMS/ADJ}$ and $V_{residual}$ are not investigated in this section as the goal of analysis is a qualitative description of very large and localized land subsidence signals, for which reference frame adjustment and GIA influence are secondary.

We selected four subsidence hot-spots that are illustrated on Figure 7. Two are large scale, well-known and well documented: The Netherlands (see e.g. Erkens et al., 2016), and the northern Italian coastal plain (see e.g. Fiaschi et al., 2018; Tos et al., 2016). The analysis is completed with two more local hot-spots: Thessaloniki in Greece and its surrounding area, reported by Stiros (2001), and the city of Schiavonela in Southern Italy and its periphery, which is here documented as a subsidence hot-spot for the first time.

In the Netherlands (panel 1 of Figure 7), V_{EGMS} estimates are lower than -1 mm/yr throughout the CFPs (light blue, blue and black pixels). Subsidence tends to diminish landward as revealed by the yellow pixels. The subsidence rate is however not uniformly distributed spatially at the scale of the Netherlands CFPs. In particular, the south-west (around Rotterdam) is characterized by several extended areas where vertical land velocities are lower than -5 mm/yr (black pixels). Subsidence in these areas is consistent with groundwater level lowering in the Holocene that caused oxidation of peat organic matter (Koster et al., 2018). In the north-east, in the area of Groningen, a large subsidence hot-spot, with velocities than can locally be lower than -5 mm/yr (even -10 mm/yr) are measured. The latter has been well documented and attributed to gas extraction (van Thienen-Visser et al., 2016). Between these two distant subsidence hot-spots, the rest of the Netherlands appears to be affected by subsidence, but to a lesser magnitude. This spatial heterogeneity of subsidence in the large CFPs of the Netherlands is very likely one of the primary reasons why the mean and median EGMS velocity over CFPs deviate (see discussion).

The northern Italian coastal plain is also well-known for being a subsidence hot-spot. Not surprisingly V_{EGMS} confirm this extended subsidence, as shown in panel 2 of Figure 7. In contrast with the Netherlands CFPs case, the subsidence pattern appears to be more uniform in the large-scale northern Italian CFPs with V_{EGMS} typically in the range -2 to -6 mm/yr. Since the CFPs in Northern Adriatic represent most of the Italian CFPs area and given the rather homogenous distribution of subsidence in them, the median and mean of CFP-scale subsidence estimates are similar (Figure 6 and Figure S5 in Supporting Information S1).

In Southern Italy, our coastal EGMS analysis identified a previously unreported coastal subsiding region in the vicinity of the city of Schiavonea and its fishing harbor (panel 3 of Figure 7). While the corresponding CFPs are not extensive, they experience strong subsidence that is mostly recorded below -5 mm/yr and co-located with exposed agriculture and urban areas. Note that the V_{GNSS} retrieved from the NGL solution for the sole GNSS station nearby (CORC station) also reveals large negative land movement velocities of nearly -5 mm/yr (calculated over less than 4 years, not shown). This very local example demonstrates the added value of EGMS to identify new and strongly subsiding coastal regions.

The last subsidence hot-spot example is Thessaloniki (panel 4 of Figure 7), which has already been documented (Raucoules et al., 2008; Stiros, 2001). In this case, a strong subsidence (i.e., V_{EGMS} lower than -5 mm/yr) is measured in two urban plots of the Aksiou delta. The Thessaloniki airport, located on the southern shore of the bay in a CFP is also experiencing strong subsidence with many EGMS pixels that observed V_{EGMS} lower than -10 mm/yr. In this area again, the subsidence pattern is highly heterogeneous.

In addition to identifying subsidence hot-spots it is also interesting to identify coastal lowlands that have limited subsidence. Both Belgium and the UK have extensive coastal lowlands that have experienced significant loss in





Figure 7. Illustration of two large scale subsidence hot-spots in the Netherlands (1), Northern Adriatic (2), and two local scale subsidence hot-spots around Schiavonea in Italy (3) and Thessaloniki in Greece (4). CFPs are shaded in gray, and European Ground Motion Service pixels and associated V_{EGMS} are colored (see Legend).

land elevation due to drainage and shallow subsidence (de la Vega-Leinert & Nicholls, 2008; Lebbe et al., 2008). For example, the Fens in eastern England has lost up to 5-m elevation since the mid 19th Century (Dawson et al., 2010). Loss of elevation is often assumed to be ongoing. However, in the period of the EGMS observations considered in this paper any subsidence appears negligible except locally (not shown) and contrasts with the observations already discussed in the Netherlands.

4.3. Exposed Assets and Population

To assess how subsidence could inflate coastal risk in low-lying coastal areas, we now relate the CFP contemporary VLM to European land cover and population, provided by CLC and Eurostat data sets, respectively (see Sections 2.2 and 2.3). As the purpose here is a quantitative assessment at pan-European scale, the analysis below relies on $\overline{V}_{EGMS/ADJ}$ (i.e., within the ITRF2014 reference frame) rather than \overline{V}_{EGMS} .

Table 3

Distribution, With Respect to Area, of Different Land Cover Types (e.g., Airports) and Exposure in Coastal Flood Plain as a Function of $\overline{V}_{EGMS/ADJ}$ or $\widetilde{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$ or $\widetilde{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$ or $\widetilde{V}_{EGMS/ADJ}$ or $\widetilde{V}_{EGMS/ADJ}$ or $\widetilde{V}_{EGMS/ADJ}$ and $\widetilde{V}_{EGMS/ADJ}$ or \widetilde{V}_{EGMS/A

]+∞;-0.5]] mm/yr]-0.5;-0	o[mm/yr]−1;−∞[mm/yr]-2;-∞	[mm/yr
VLM in CFP exposure	$\overline{V}_{EGMS/ADJ}$	$\widetilde{V}_{EGMS/ADJ}$	$\overline{V}_{EGMS/ADJ}$	$\widetilde{V}_{EGMS/ADJ}$	$\overline{V}_{EGMS/ADJ}$	$\widetilde{V}_{EGMS/ADJ}$	$\overline{V}_{EGMS/ADJ}$	$\widetilde{V}_{EGMS/ADJ}$
Continuous urban fabric	25%	38%	73%	60%	31%	12%	5%	1%
Discontinuous urban fabric	14%	27%	85%	72%	50%	10%	7%	2%
Industrial or commercial units	15%	28%	84%	70%	49%	13%	8%	3%
Road and rail networks and associated land	8%	15%	92%	85%	52%	17%	10%	3%
Port areas	22%	31%	76%	67%	42%	22%	10%	4%
Airports	28%	41%	68%	55%	45%	20%	13%	5%
Inland marshes	15%	22%	81%	74%	66%	20%	17%	2%
Salt marshes	27%	43%	68%	52%	33%	17%	10%	6%
Salinas	12%	35%	79%	56%	44%	24%	14%	12%
Population	22%	27%	74%	69%	49%	16%	7%	3%

Note. This reads for example, as follows: 50% of the area occupied by discontinuous urban fabric in CFPs over Europe is located in CFPs with $\overline{V}_{EGMS/ADJ}$ lower than -1 mm/yr. Note that the sum over the ranges]+ ∞ ;-0.5] and]-0.5;- ∞ [mm/yr does not reach 100% since even if all CFPs have CLC and population information, not all have EGMS pixels (see e.g. Table 1).

Table 3 relates the land cover to the state of $\overline{V}_{EGMS/ADJ}$ in CFPs in all Europe. The analysis is restricted to a subset of 9 classes (among 44) of CLC those land cover generally include highly exposed anthropogenic (i.e., cities, airports, commercial centers,...) or coastal natural (i.e., marshes) assets. The results reveal that a large majority (more than 70%) of assets within CFPs in Europe are located in CFPs with $\overline{V}_{EGMS/ADJ}$ lower than -0.5 mm/yr. For instance, 92% of the area occupied by transport networks is located in CFPs with $\overline{V}_{EGMS/ADJ}$ lower than -0.5 mm/ yr. The ratio of exposed assets is still high when considering CFPs with $\overline{V}_{EGMS/ADJ}$ lower than -1 mm/yr: this is for example, the case for nearly half of discontinuous urban fabric, industries, transport networks or inland marshes located in CFPs. Consistently, half of the European population living in the CFP (which represents ~16 million inhabitants in the Eurostat data set) is also located in CFPs with $\overline{V}_{EGMS/ADJ}$ lower than -1 mm/yr. Finally, near 10% of the CFPs area covered by exposed assets is located in CFPs with $\overline{V}_{EGMS/ADJ}$ lower than -2 mm/yr.

Although Table 3 provides an overall appraisal of the exposure to subsidence at the CFP scale, the analysis can be further refined by calculating within each CFP, the $\overline{V}_{EGMS/ADJ}$ (i.e., the average) only over pixels associated with each CLC class. The refined analysis is illustrated in Figure 8 for the CLC class "urban fabric" (both continuous and discontinuous combined). Here, for each pixel of CFPs containing the urban fabric class, we select the colocated $V_{EGMS/ADJ}$, calculate the average within each CFP (i.e., $\overline{V}_{EGMS/ADJ}$) and draw the corresponding European distribution per CFP (Figure 8). This concerns 4,641 CFPs among ~41,000. It reveals a bimodal distribution with a first peak slightly negative (near -1 mm/yr) and a second much smaller peak toward large positive values (~9 mm/yr). This second peak is related to the influence of GIA in Northern Europe. This refined analysis has been repeated for the CLC classes "Harbor" and "Airport" and for the median, first and third quartiles of $V_{EGMS/ADJ}$ to reflect the spread of the estimates. The main results are summarized in Table 4.

The mean of $\overline{V}_{EGMS/ADJ}$ co-located with urban fabric (i.e., mean of the distribution shown on Figure 8) is 0.57 mm/yr when considering the whole Europe (Table 4). If now the mean is weighted by either the population or the area of the corresponding land cover type within each CFP, we find a similar (and consistent) subsiding velocity of urban fabric areas of nearly 0.5 mm/yr. If the CFPs located in a region where GIA triggers a land uplift are discarded, the mean $\overline{V}_{EGMS/ADJ}$ for urban fabric experience stronger subsidence (0.8 mm/yr for area- and population-weighted estimates). In other words, this analysis shows that the European population settled in coastal flood prone areas lives on average where VLM of nearly -1 mm/yr is measured. For airports and harbors, subsidence rates are even stronger (-1 to -1.5 mm/yr). For ports, which are directly connected to the coast and for which sea level is a critical operating parameter, this means that they currently experience an additional relative SLR rate of 1 mm/yr on average. For some ports, this can be even larger. Finally, quartiles estimates display mostly consistent signs with the mean estimates (Table 4). Hence, despite the fact that vertical land







Figure 8. Distribution of $\overline{V}_{EGMS/ADJ}$ co-located with urban fabric (i.e., continuous and discontinuous) land cover in European CFPs.

velocities estimates vary spatially within CFPs, the mean estimate is relevant to capture the global land motion behavior of CFPs and associated land cover types.

5. Discussion

5.1. Significance of Our Main Findings

In this paper, we present for the first time a pan-European analysis of contemporary VLM and subsidence in low-lying flood prone coastal zones using the EGMS product. First, we perform a comparison between EGMS Ortho (Level 3) vertical velocity estimates and GNSS stations vertical velocity estimates from two different solutions (ULR7A & NGL). This comparison reveals that the geodetic reference frame used to calibrate EGMS can strongly affect the vertical land velocity estimates and needs to be accounted for carefully, especially for coastal applications where the need of sub-mm/yr precision in VLM is required for comparison with SLR changes (as opposed e.g. to landslides or earthquakes that are also characterized by EGMS). As coastal sea-level applications require precision and stability of positioning of the geodetic reference frame, VLM are generally expressed within ITRF (Wöppelmann & Marcos, 2016). We could hence determine and apply a simple polynomial function depending on latitude to transform EGMS VLM velocities from ETRF2000 (used to calibrate EGMS) to ITRF2014 in order to compensate the identified bias. Overall this leads to a reduction of the subsiding trend which amplifies northward. The adjustment we applied is

empirical, based on vertical velocity estimates from the same GNSS network of stations (EPN) expressed with respect to these two different reference frames. As shown in Figure 3b, the fit of our transfer function is acceptable given that EGMS pixels are provided with a precision of 1 mm/yr.

Based on our adjusted estimates, we found that 50% of the European CFPs spatial extent is, on average, experiencing subsidence at a rate stronger than 1 mm/yr and 10% are uplifting at a larger rate than 0.5 mm/yr (Figure 4). If the GIA contribution is removed, it reduces the CFP area uplifting to 5% and the CFP area subsiding at rates lower than 1 mm/yr to slightly less than 20% (on average). This shows the substantial first-order effect of GIA in both land uplift and subsidence trends across Europe. However, the detailed results depend on the GIA model employed. Here we used the Caron et al. (2018) model, which has been successfully employed recently to close the basin-scale 20th century sea level budget and which appeared within the confidence intervals of the recent ICE-6G_C(VM5a) GIA model (Frederikse et al., 2020; Peltier et al., 2018). Nonetheless, Vermeersen et al. (2018) found that discrepancies between GIA models obtained from different methods and source data

Table 4Mean of $\overline{V}_{EGMS/ADJ}$, $\widetilde{V}_{EGMS/ADJ}$, First and Third Quartiles of $V_{EGMS/ADJ}$ for Urban Fabric, Harbor, and Airport Land Cover

		All Europe	CFP with no uplift GIA
Urban fabric (4641)	No weight	0.57 mm/yr (0.40/0.63/0.81)	-0.57 mm/yr (-0.74/-0.51/-0.35)
	Population weighted	-0.52 mm/yr (-0.82/-0.35/-0.04)	-0.86 mm/yr (-1.17/-0.68/-0.35)
	Area weighted	-0.48 mm/yr (-0.80/-0.29/0.04)	-0.79 mm/yr (-1.10/-0.59/-0.25)
Harbor (1116)	No weight	-0.51 mm/yr (-0.81/-0.39/-0.09)	-0.95 mm/yr (-1.23/-0.86/-0.58)
	Area weighted	-1.01 mm/yr (-1.51/-0.64/-0.11)	-1.46 mm/yr (-1.91/-1.12/-0.60)
Airport (188)	No weight	0.52 mm/yr (0.26/0.66/0.94)	-0.66 mm/yr (-0.94/-0.52/-0.24)
	Area weighted	-0.55 mm/yr (-0.93/-0.31/0.11)	-1.07 mm/yr (-1.47/-0.81/-0.37)

Note. Averages are computed either equally over all CFPs (i.e., no weight) or, in the case of urban fabric, are also weighted with respect to the population in CFPs (population weighted) and area of the corresponding land cover type (area weighted) within each CFP. Numbers in brackets indicate the total number of CFPs concerned by the CLC class. The mean of the first quartile, the median and the third quartile of $V_{EGMS/ADJ}$ are noted in brackets and separated by slashes.

(e.g., data-driven model (Simon et al., 2018), regional GIA model (Bradley et al., 2011) or Ice-sheet history-based 3D-ice sheet simulations (Kuchar et al., 2012)) can significantly affect relative sea level regionally (see e.g. Figure 7 in Vermeersen et al. (2018) or Bradley et al. (2023)). Given the importance of GIA for VLM in Europe, uncertainties related to the GIA model employed need to be analyzed in more depth in future research work.

Comparing Figure 6 (based on mean estimates per CFP) and the Figure S5 in Supporting Information S1 (based on median estimates per CFP), we find consistent results in Northern and Southern European countries, but quite a substantial difference in the Netherlands, Germany and France. Indeed, for the later countries, the median estimates trends are less strong (negatively) than the mean estimates. The panel 1 of Figure 7, which details the Netherlands distribution, shows that within one very expanded CFP, the VLM spatial distribution can be very heterogeneous, suggesting various underlying physical processes (e.g., Erkens et al., 2016). While this heterogeneity of the VLM distribution within one CFP can explain why median and mean estimates are different, it also highlights that for more detailed analysis, these two summary statistics may not always provide enough information to describe the state of VLM in coastal plains, and a complete statistical distribution may be required.

Beside the use of EGMS for the European assessment (Section 4.1), we also demonstrate how to take advantage of the pixel scale estimate (Sections 4.2 and 4.3). We are able to identify a subsiding coastal area (i.e., Schiavonea) that was not previously recognized or considered in previous pan-European coastal impact assessments. The averaged results combined with the CLC data set allow drawing a significant overall behavior. Indeed, we find that urban areas and ~16 millions residents are experiencing almost a 1 mm/yr subsidence on average (if we discard the uplifting regions due to GIA). For coastal airports and for harbors, the average land motion drops to a 1-1.5 mm/yr subsidence leading to a substantial relative SLR enhancement locally. This may have direct consequences on emerging risks and 21st century SLR adaptation planning and need to be considered to update previous coastal flooding risk assessments (e.g., Hauer et al., 2021).

We also note that while in this work we used EGMS information from a relative SLR perspective, further studies dedicated to local coastal impacts could take further advantage of the gridded VLM information. Indeed, spatial VLM could be used to adjust the elevation of DEMs and DTMs and ultimately to simulate flooding and erosion using coastal impact models (e.g., Miller & Shirzaei, 2021; Minderhoud et al., 2020; Ohenhen et al., 2024).

5.2. Broader Implications and Research Needs

As we move from understanding the magnitudes of SLR and the potential for climate mitigation to coastal adaptation, the importance of high quality relative SLR scenarios becomes increasingly apparent (e.g., Bednar-Friedl et al., 2022). Yet, the quality of coastal relative sea-level projections is limited by the lack of information on VLM (Slangen et al., 2023): this is acknowledged in the IPCC AR6, which assigns "*low to medium confidence*" to VLMs projections presented in the report (Fox-Kemper et al., 2021). In areas affected by subsidence, overlooking VLMs could significantly underestimate the risks and reduce the time available for adaptation as critical thresholds such as seawall heights or the capacity of flood expansion zones in estuaries will be exceeded earlier than projected.

Services such as the EGMS offer an obvious response to this need for improved information on present-day and future VLM. Importantly, such services do not only integrate information from the SAR Sentinel satellites, but also include information from the permanent GNSS networks to link VLMs tiles obtained from SAR interferometry. This, in turn, builds upon three decades of research in space geodesy (e.g., Bruyninx, 2004; Ferretti et al., 2001). For example, our study provides, to our knowledge, a first practical demonstration of the crucial role played by the terrestrial reference frame in which the InSAR results are expressed (e.g., Wöppelmann & Marcos, 2016). The type of analysis illustrated in Figure 3 is crucial to all users of broad scale land motion observations based on InSAR, whether in Europe or in other regions.

Our results show that when developing climate services to assess coastal future flood risks in Europe, estimates of uplift and subsidence were essential. These needs would be stronger in other regions where subsidence is a more important process, most especially east, south-east and south Asia (Nicholls, Lincke, et al., 2021). In these regions, research studies may not be sufficient and operational services similar to EGMS may be useful in order to provide stakeholders with validated and standardized products. We note that such services have emerged nationally before EGMS was opened as a Europe-wide service: the first Ground Motion Service (GMS) covered Italy (Costantini et al., 2017). This was followed by Norway in 2018, Germany in 2019 and then Denmark and the

Netherlands. All other countries can consider similar services owing to the almost global coverage of Sentinel-1 constellation of satellite and its past and planned continuity of service.

It is important to note that the subsidence estimates provided here are based on a short period of observations - the five-year EGMS period of 2015–2021. As EGMS is updated on an annual basis, repeating this analysis should allow updating of the VLM estimates provided here and monitoring their continuity in time. This also raises the potential to actively monitor VLM at broad scales, such as Europe, and identify any new or enhanced subsidence hot-spots that might emerge due to changing land use and groundwater exploitation. SAR interferometry, on which EGMS relies, has limitations in vegetated areas due to the lack of backscatterers on the ground. From the perspective of the application presented in this paper, this is not a major limitation as relative SLR projections are most useful in areas with high exposure, that is, in cities and settlements. In such areas InSAR performs well.

Given the huge amount of EGMS data available at the European scale, we had to restrict our analysis to the mean velocity estimates provided at each pixel. For more detailed studies in space (e.g., at the city scale), considering the full time series could be relevant, for example, to test the stationarity of the trends and better quantify their robustness. The latter point also raises the important question of how to develop VLM projections to complement climate-induced sea-level projections: how can we best use EGMS estimates to infer future VLM and especially subsidence? This is a topic for further research.

6. Conclusions

In this paper we present a pan-European assessment of contemporary VLMs in the coastal flood prone areas based on EGMS InSAR-based products. Our estimates show that near half of the total CFP area is on average subsiding at a rate faster than 1 mm/yr. We find that urban zones and population in flood-prone areas, in European regions not subjected to GIA-induced land uplift (i.e., Fennoscandia), currently experience a subsidence of nearly 1 mm/ yr. For harbors and airports located in flood-prone areas, the average coastal subsidence is more than 1 mm/yr. This locally enhances the relative SLR rate by about one third (in comparison with the ongoing climate-induced regional mean sea level in Europe which has risen by an average of 2–4 mm/year over the past 30 years; https:// climate.copernicus.eu/climate-indicators/sea-level) and demonstrates the significance of subsidence in Europe when considering coastal flood risk.

For coastal adaptation practitioners, the next question that arises is to understand to what extent observed VLMs can be extrapolated into the future. The question is important from a policy perspective as adaptation practitioners across the world are increasingly using and refining sea-level scenarios for risk assessment and adaptation planning (Hirschfeld et al., 2023). Another question is how to support decision making at the level of specific infrastructure, as the EGMS shows that some dikes and other coastal infrastructures are subject to subsidence, possibly in part due to their own weight. This will require further research and ensuring the continuity of observations in EGMS to confirm trends and identify potential non-linearities.

Notwithstanding these challenges, our study shows that there is a potential for a service for coastal adaptation practitioners, positioned downstream of EGMS or similar services in other regions worldwide. EGMS is a service of the "land" component of the European Earth Observation program Copernicus, but delivering actionable information to coastal users concerned with relative SLR requires additional analysis, as shown in this paper. Similar services can be developed in other regions of the world owing to the almost global coverage of the world's coasts by the Sentinel 1 constellation of satellites. Given the challenges raised by SLR, it would be important to make sure that such operational services are further tailored to the needs of coastal practitioners concerned with understanding risks and planning for adaptation.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

EGMS Ortho products first release (2015–2021) is obtained from European Environment Agency (2023). The ULR-repro3 GNSS products are available from Gravelle et al. (2022). The GNSS NGL14 solution (Blewitt et al., 2018) used in this study is distributed by SONEL at the landing page (https://www.sonel.org/-Vertical-land-



coastal flood plains are provided in Thiéblemont (2024).

This work is funded by CoCliCo and GSEU projects. CoCliCo project has received funding from the European Union's Horizon 2020 research and innovation programme under Grant 101003598. GSEU has received funding from the European Union's Horizon Europe research & innovation programme under Grant 101075609 (HORIZON-CL5-2021-D3-02). We thank the European Copernicus Land Monitoring Service and colleagues who produced and released the EGMS service and CLC. We thank the SONEL platform. We thank researchers Jonathan Chenal, involved in CoCliCo, for the fruitful discussions on geodesy and Aurélie Maspataud for her support as coleader of the WP5 of the Horizon GSEU program. Alexandra Toimil acknowledges the financial support from the Ministerio de Ciencia e Innovación through the Ramon y Cajal Programme (RYC2021-030873-I with funding from MCIN/AEI and NextGenerationEU/PRTR).

References

Altamimi, Z., Métivier, L., Rebischung, P., Rouby, H., & Collilieux, X. (2017). ITRF2014 plate motion model. Geophysical Journal International, 209(3), 1906–1912. https://doi.org/10.1093/gji/ggx136

movements-.html?lang=en#id498). EUREF EPN D2150 densification GNSS solution was obtained from EUREF (https://epnd.sgo-penc.hu/downloads/). The results from the GIA model were downloaded from Fred-

erikse, Caron, and Landerer (2019). CORINE Land Cover raster is obtained from CLC raster is obtained from European Environment Agency (2020). European population gridded data set from 2021 is obtained from European (2023). The Copernicus Global Digital Elevation Model at 30-m resolution used to define European coastal flood plains from European Space Agency, Sinergise (2021). Vertical land motion estimates in European

- Antonioli, F., Ferranti, L., Fontana, A., Amorosi, A., Bondesan, A., Braitenberg, C., et al. (2009). Holocene relative sea-level changes and vertical movements along the Italian and Istrian coastlines. *Quaternary International*, 206(1–2), 102–133. https://doi.org/10.1016/j.quaint.2008.11.008
- Ballu, V., Gravelle, M., Wöppelmann, G., de Viron, O., Rebischung, P., Becker, M., & Sakic, P. (2019). Vertical land motion in the Southwest and Central Pacific from available GNSS solutions and implications for relative sea levels. *Geophysical Journal International*, 218(3), 1537–1551. https://doi.org/10.1093/gji/ggz247
- Bednar-Friedl, B., Biesbroek, R., Schmidt, D. N., Alexander, P., Børsheim, K. Y., Carnicer, J., et al. (2022). Europe. In H.-O. Pörtner, D. C. Roberts, M. Tignor, E. S. Poloczanska, K. Mintenbeck, A. Alegría, et al. (Eds.), *Climate change 2022: Impacts, adaptation and vulnerability. Contribution of working group II to the sixth assessment report of the intergovernmental panel on climate change* (pp. 1817–1927). Cambridge University Press. https://doi.org/10.1017/9781009325844.015
- Blewitt, G., Hammond, W. C., & Kreemer, C. (2018). Harnessing the GPS data explosion for interdisciplinary science. *Eos*, 99. https://doi.org/10. 1029/2018EO104623
- Blewitt, G., Kreemer, C., Hammond, W. C., & Gazeaux, J. (2016). MIDAS robust trend estimator for accurate GPS station velocities without step detection. Journal of Geophysical Research, 121(3), 2054–2068. https://doi.org/10.1002/2015JB012552
- Bossard, M., Feranec, J., & Otahel, J. (2000). CORINE land cover technical guide Addendum 2000. Technical report no 40. European Environmental Agency. Retrieved from https://www.eea.europa.eu/publications/tech40add/at_download/file
- Bradley, S. L., Ely, J. C., Clark, C. D., Edwards, R. J., & Shennan, I. (2023). Reconstruction of the palaeo-sea level of Britain and Ireland arising from empirical constraints of ice extent: Implications for regional sea level forecasts and North American ice sheet volume. *Journal of Quaternary Science*, 38(6), 791–805. https://doi.org/10.1002/jqs.3523
- Bradley, S. L., Milne, G. A., Shennan, I., & Edwards, R. (2011). An improved Glacial Isostatic Adjustment model for the British Isles. Journal of Quaternary Science, 26(5), 541–552. https://doi.org/10.1002/jqs.1481
- Bruyninx, C. (2004). The EUREF Permanent Network: A multi-disciplinary network serving surveyors as well as scientists. *Geoinformatics*, 7, 32–35.
- Bruyninx, C., Legrand, J., Fabian, A., & Pottiaux, E. (2019). GNSS metadata and data validation in the EUREF Permanent Network. GPS Solutions, 23(4), 106. https://doi.org/10.1007/s10291-019-0880-9
- Burkett, V. B., Zilkowski, D. B., & Hart, D. A. (2003). Sea level rise and subsidence: Implications for flooding in New Orleans, Louisiana. In K. R. Prince & D. L. Galloway (Eds.), Proc. USGS aquifer mechanics and subsidence interest group conference (pp. 63–70). US Geological Survey.
- Cao, A., Esteban, M., Valenzuela, V. P. B., Onuki, M., Takagi, H., Thao, N. D., & Tsuchiya, N. (2021). Future of Asian Deltaic Megacities under sea level rise and land subsidence: Current adaptation pathways for Tokyo, Jakarta, Manila, and Ho Chi Minh City. Current Opinion in Environmental Sustainability, 50, 87–97. https://doi.org/10.1016/j.cosust.2021.02.010
- Capes, R., & Passera, E. (2023). Product description and format specification. EGMS v2.0. Retrieved from https://land.copernicus.eu/en/technicallibrary/egms-product-description-document/@@download/file
- Carbognin, L., Teatini, P., & Tosi, L. (2004). Eustacy and land subsidence in the Venice Lagoon at the beginning of the new millennium. Journal of Marine Systems, 51(1–4), 345–353. https://doi.org/10.1016/j.jmarsys.2004.05.021
- Carminati, E., Doglioni, C., & Scrocca, D. (2003). Apennines subduction-related subsidence of Venice (Italy). *Geophysical Research Letters*, 30(13), 1717. https://doi.org/10.1029/2003GL017001
- Caron, L., Ivins, E. R., Larour, E., Adhikari, S., Nilsson, J., & Blewitt, G. (2018). GIA model statistics for GRACE hydrology, cryosphere, and ocean science. *Geophysical Research Letters*, 45(5), 2203–2212. https://doi.org/10.1002/2017GL076644
- Cavalié, O., Cappa, F., & Pinel-Puysségur, B. (2023). Three decades of coastal subsidence in the slow-moving Nice Côte d'Azur Airport area (France) revealed by InSAR (interferometric synthetic-aperture radar): Insights into the deformation mechanism. *Natural Hazards and Earth System Sciences*, 23(10), 3235–3246. https://doi.org/10.5194/nhess-23-3235-2023
- Cooley, S., Schoeman, D., Bopp, L., Boyd, P., Donner, S., Ito, S.-i., et al. (2022). Oceans and coastal ecosystems and their services. In *IPCC AR6 WGII* (pp. 379–550). Cambridge University Press. https://doi.org/10.1017/9781009325844.005
- Costantini, M., Ferretti, A., Minati, F., Falco, S., Trillo, F., Colombo, D., et al. (2017). Analysis of surface deformations over the whole Italian territory by interferometric processing of ERS, Envisat and COSMO-SkyMed radar data. *Remote Sensing of Environment*, 202, 250–275. https://doi.org/10.1016/j.rse.2017.07.017
- Costantini, M., Minati, F., Trillo, F., Ferretti, A., Novali, F., Passera, E., et al. (2021). European ground motion service (EGMS). In *IEEE international geoscience and remote sensing symposium IGARSS, Brussels, Belgium* (pp. 3293–3296). https://doi.org/10.1109/IGARSS47720. 2021.9553562
- Crosetto, M., Monserrat, O., Cuevas-González, M., Devanthéry, N., & Crippa, B. (2016). Persistent scatterer interferometry: A review. ISPRS Journal of Photogrammetry and Remote Sensing, 115, 78–89. https://doi.org/10.1016/j.isprsjprs.2015.10.011
- Crosetto, M., & Solari, L. (2023). Satellite interferometry data interpretation and exploitation: Case studies from the European Ground Motion Service (EGMS). Elsevier.
- Crosetto, M., Solari, L., Balasis-Levinsen, J., Bateson, L., Casagli, N., Frei, M., et al. (2021). Deformation monitoring at European scale: The copernicus ground motion service. *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, 43, 141–146. https://doi.org/10.5194/isprs-archives-xliii-b3-2021-141-2021
- Crosetto, M., Solari, L., Mróz, M., Balasis-Levinsen, J., Casagli, N., Frei, M., et al. (2020). The evolution of wide-area DInSAR: From regional and national services to the European ground motion service. *Remote Sensing*, *12*, 2043. https://doi.org/10.3390/rs12122043

- Cuffaro, M., Riguzzi, F., Scrocca, D., Antonioli, F., Carminati, E., Livani, M., & Doglioni, C. (2010). On the geodynamics of the northern Adriatic plate. *Rendiconti Lincei*, 21(S1), 253–279. https://doi.org/10.1007/s12210-010-0098-9
- Dawson, Q., Kechavarzi, C., Leeds-Harrison, P. B., & Burton, R. G. O. (2010). Subsidence and degradation of agricultural peatlands in the Fenlands of Norfolk, UK. *Geoderma*, 154(3–4), 181–187. https://doi.org/10.1016/j.geoderma.2009.09.017
- de la Vega-Leinert, A. C., & Nicholls, R. J. (2008). Potential implications of sea-level rise for Great Britain. Journal of Coastal Research, 24(2), 342–357.
- Erban, L. E., Gorelick, S. M., & Zebker, H. A. (2014). Groundwater extraction, land subsidence, and sea-level rise in the Mekong Delta, Vietnam. Environmental Research Letters, 9(8), 084010. https://doi.org/10.1088/1748-9326/9/8/084010
- Erkens, G., van der Meulen, M. J., & Middelkoop, H. (2016). Double trouble: Subsidence and CO2 respiration due to 1,000 years of Dutch coastal peatlands cultivation. *Hydrogeology Journal*, 24(3), 551–568. https://doi.org/10.1007/s10040-016-1380-4
- Esteban, M., Takagi, H., Jamero, L., Chadwick, C., Avelino, J. E., Mikami, T., et al. (2020). Adaptation to sea level rise: Learning from present examples of land subsidence. Ocean & Coastal Management, 189, 104852. https://doi.org/10.1016/j.ocecoaman.2019.104852
- European Environment Agency. (2020). CORINE land cover 2018 (raster 100 m), Europe, 6-yearly Version 2020_20u1, May 2020 [Dataset]. https://doi.org/10.2909/960998c1-1870-4e82-8051-6485205ebbac
- European Environment Agency. (2023). European Ground Motion Service: Ortho Vertical component 2015-2021 (vector) [Dataset], Europe, yearly, Feb. 2023. https://doi.org/10.2909/0a94b5d4-b414-4f2b-a6be-eea73094a0f5
- European Space Agency, Sinergise. (2021). Copernicus Global Digital Elevation Model [Dataset]. Distributed by OpenTopography. https://doi. org/10.5069/G9028POB
- Eurostat. (2023). Eurostat 2021 grid census [Dataset], EUROSTAT, the statistical office of the European Union, Luxembourg, Luxembourg. https://ec.europa.eu/eurostat/web/gisco/geodata/reference-data/population-distribution-demography/geostat
- Farolfi, G., Bianchini, S., & Casagli, N. (2019). Integration of GNSS and satellite InSAR data: Derivation of fine-scale vertical surface motion maps of Po Plain, Northern Apennines, and Southern Alps, Italy. *IEEE Transactions on Geoscience and Remote Sensing*, 57(1), 319–328. https://doi.org/10.1109/TGRS.2018.2854371
- Feranec, J., Soukup, T., Hazeu, G., & Jaffrain, G. (Eds.). (2016). European landscape dynamics. Corine land cover data (pp. 9–14). CRC-Press. Ferretti, A., Passera, E., & Capes, R. (2023). Algorithm theoretical basis document. EGMS V3.0. Retrieved from https://land.copernicus.eu/en/ technical-library/eems-algorithm-theoretical-basis-document/@@download/file
- Ferretti, A., Prati, C., & Rocca, F. (2001). Permanent scatterers in SAR interferometry. IEEE Transactions on Geoscience and Remote Sensing, 39(1), 8–20. https://doi.org/10.1109/36.898661
- Fiaschi, S., Fabris, M., Floris, M., & Achilli, V. (2018). Estimation of land subsidence in deltaic areas through differential SAR interferometry: The Po River Delta case study (Northeast Italy). *International Journal of Remote Sensing*, 39(23), 8724–8745. https://doi.org/10.1080/ 01431161.2018.1490977
- Fokker, P. A., Gunnink, J. L., Koster, K., & de Lange, G. (2019). Disentangling and parameterizing shallow sources of subsidence: Application to a reclaimed coastal area, Flevoland, The Netherlands. *Journal of Geophysical Research: Earth Surface*, 124(5), 1099–1117. https://doi.org/10. 1029/2018jf004975
- Fox-Kemper, B., Hewitt, H. T., Xiao, C., Aðalgeirsdóttir, G., Drijfhout, S. S., Edwards, T. L., et al. (2021). Ocean, Cryosphere and sea level change. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), Climate change 2021: The physical science basis. Contribution of working Group I to the sixth assessment report of the intergovernmental panel on climate change. Cambridge University Press.
- Frederikse, T., Caron, L., & Landerer, F. (2019b). The imprints of contemporary mass redistribution on regional sea level and vertical land motion observations [Dataset]. https://doi.org/10.5281/zenodo.3485577
- Frederikse, T., Landerer, F., Caron, L., Adhikari, S., Parkes, D., Humphrey, V. W., et al. (2020). The causes of sea-level rise since 1900. Nature, 584(7821), 393–397. https://doi.org/10.1038/s41586-020-2591-3
- Frederikse, T., Landerer, F. W., & Caron, L. (2019a). The imprints of contemporary mass redistribution on local sea level and vertical land motion observations. Solid Earth, 10(6), 1971–1987. https://doi.org/10.5194/se-10-1971-2019
- Galloway, D. L., & Burbey, T. J. (2011). Review: Regional land subsidence accompanying groundwater extraction. *Hydrogeology Journal*, 19(8), 1459–1486. https://doi.org/10.1007/s10040-011-0775-5
- Gatto, P., & Carbognin, L. (1981). The Lagoon of Venice: Natural environmental trend and man-induced modification/La Lagune de Venise: ''évolution naturelle et les modifications humaines. *Hydrological Sciences Journal*, 26(4), 379–391. https://doi.org/10.1080/ 02626668109490902
- Giosan, L., Syvitski, J. P. M., Constantinescu, S., & Day, J. (2014). Protect the world's deltas. Nature, 516, 5-7.
- Gravelle, M., Wöppelmann, G., Gobron, K., Altamimi, Z., Guichard, M., Herring, T., & Rebischung, P. (2023). The ULR-repro3 GPS data reanalysis and its estimates of vertical land motion at tide gauges for sea level science. *Earth System Science Data*, *15*(1), 497–509. https://doi. org/10.5194/essd-15-497-2023
- Gravelle, M., Wöppelmann, G. M., Gobron, K., Altamimi, Z., Guichard, M., Herring, T., & Rebischung, P. (2022). The ULR-repro3 GPS data reanalysis solution (aka ULR7a), SONEL Data Center [Dataset]. https://doi.org/10.26166/sonel_ulr7a
- Hasan, M. F., Smith, R., Vajedian, S., Pommerenke, R., & Majumdar, S. (2023). Global land subsidence mapping reveals widespread loss of aquifer storage capacity. *Nature Communications*, 14(1), 6180. https://doi.org/10.1038/s41467-023-41933-z
- Hauer, M. E., Hardy, D., Kulp, S. A., Mueller, V., Wrathall, D. J., & Clark, P. U. (2021). Assessing population exposure to coastal flooding due to sea level rise. *Nature Communications*, 12(1), 6900. https://doi.org/10.1038/s41467-021-27260-1
- Heflin, M., Donnellan, A., Parker, J., Lyzenga, G., Moore, A., Ludwig, L. G., et al. (2020). Automated estimation and tools to extract positions, velocities, breaks, and seasonal terms from daily GNSS measurements: Illuminating nonlinear Salton Trough deformation. *Earth and Space Science*, 7, e2019EA000644. https://doi.org/10.1029/2019EA000644
- Hinkel, J., Lincke, D., Vafeidis, A. T., Perrette, M., Nicholls, R. J., Tol, R. S. J., et al. (2014). Future coastal flood damage and adaptation costs. Proceedings of the National Academy of Sciences, 111(9), 3292–3297. https://doi.org/10.1073/pnas.1222469111
- Hinkel, J., Nicholls, R. J., Tol, R. S. J., Wang, Z. B., Hamilton, J. M., Boot, G., et al. (2013). A global analysis of erosion of sandy beaches and sealevel rise: An application of DIVA. *Global and Planetary Change*, 111, 150–158. https://doi.org/10.1016/j.gloplacha.2013.09.002
- Hirschfeld, D., Behar, D., Nicholls, R. J., Cahill, N., James, T., Horton, B. P., et al. (2023). Global survey shows planners use widely varying sealevel rise projections for coastal adaptation. *Communications Earth & Environment*, 4(1), 102. https://doi.org/10.1038/s43247-023-00703-x Kent, D. V., Rio, D., Massari, F., Kukla, G., & Lanci, L. (2002). Emergence of Venice during the Pleistocene. *Quaternary Science Reviews*,
 - 21(14–15), 1719–1727. https://doi.org/10.1016/S0277-3791.dl/D0100153-6
- Kenyeres, A., Bellet, J. G., Bruyninx, C., Caporali, A., de Doncker, F., Droscak, B., et al. (2019). Regional integration of long-term national dense GNSS network solutions. GPS Solutions, 23(4), 122. https://doi.org/10.1007/s10291-019-0902-7

Licens

- Koster, K., De Lange, G., Harting, R., De Heer, E., & Middelkoop, H. (2018). Characterizing void ratio and compressibility of Holocene peat with CPT for assessing coastal–deltaic subsidence. *The Quarterly Journal of Engineering Geology and Hydrogeology*, 51(2), 210–218. https://doi.org/10.1144/qiegh2017-120
- Kosztra, B., Büttner, G., Hazeu, G., & Arnold, S. (2017). Updated CLC illustrated nomenclature guidelines. European Topic Centre on Urban, Land and Soil Systems. Environment Agency Austria.
- Krien, Y., Karpytchev, M., Ballu, V., Becker, M., Grall, C., Goodbred, S., et al. (2019). Present-day subsidence in the Ganges-Brahmaputra-Meghna Delta: Eastern amplification of the Holocene sediment loading contribution. *Geophysical Research Letters*, 46(19), 10764–10772. https://doi.org/10.1029/2019GL083601
- Kuchar, J., Milne, G., Hubbard, A., Patton, H., Bradley, S., Shennan, I., & Edwards, R. (2012). Evaluation of a numerical model of the British-Irish ice sheet using relative sea-level data: Implications for the interpretation of trimline observations. *Journal of Quaternary Science*, 27(6), 597–605. https://doi.org/10.1002/jqs.2552
- Lambeck, K., Smither, C., & Johnston, P. (1998). Sea-level change, glacial rebound and mantle viscosity for northern Europe. *Geophysical Journal International*, 134(1), 102–144. https://doi.org/10.1046/j.1365-246x.1998.00541.x
- Larsen, Y., Marinkovic, P., Kenyeres, A., & Toth, S. (2023). GNSS calibration report. EGMS v1.0. Retrieved from https://land.copernicus.eu/en/technical-library/gnss-calibration-report/@@download/file
- Lebbe, L., Van Meir, N., & Viaene, P. (2008). Potential implications of sea-level rise for Belgium. Journal of Coastal Research, 24(2), 358–366. https://doi.org/10.2112/07a-0009.1
- Leeper, R., Rhodes, B., Kirby, M., Scharer, K., Carlin, J., Hemphill-Haley, E., et al. (2017). Evidence for coseismic subsidence events in a southern California coastal saltmarsh. *Scientific Reports*, 7(1), 44615. https://doi.org/10.1038/srep44615
- Lincke, D., & Hinkel, J. (2021). Coastal migration due to 21st century sea-level rise. Earth's Future, 9(5), e2020EF001965. https://doi.org/10. 1029/2020EF001965
- Lincke, D., Hinkel, J., Mengel, M., & Nicholls, R. J. (2022). Understanding the drivers of coastal flood exposure and risk from 1860 to 2100. *Earth's Future*, 10(12), e2021EF002584. https://doi.org/10.1029/2021EF002584
- Lionello, P., Nicholls, R. J., Umgiesser, G., & Zanchettin, D. (2021). Venice flooding and sea level: Past evolution, present issues, and future projections (introduction to the special issue). Natural Hazards and Earth System Sciences, 21(8), 2633–2641. https://doi.org/10.5194/nhess-21-2633-2021
- Loschetter, A., Rohmer, J., Raucoules, D., & De Michele, M. (2015). Sizing a geodetic network for risk-oriented monitoring of surface deformations induced by CO2 injection: Experience feedback with InSAR data collected at In-Salah, Algeria. *International Journal of Greenhouse Gas Control*, 42, 571–582. https://doi.org/10.1016/j.ijggc.2015.09.011
- Lutz, W., Amran, G., Bélanger, A., Conte, A., Gailey, N., Ghio, D., et al. (2019). Demographic scenarios for the EU: Migration, population and education. Publications Office of the European Union.
- Männel, B., Schöne, T., Bradke, M., & Schuh, H. (2022). Vertical land motion at tide gauges observed by GNSS: A new GFZ-TIGA solution. In International association of geodesy symposia. Springer. https://doi.org/10.1007/1345_2022_150
- Meckel, T. A., ten Brink, U. S., & Williams, S. J. (2007). Sediment compaction rates and subsidence in deltaic plains: Numerical constraints and stratigraphic influences. *Basin Research*, 19(1), 19–31. https://doi.org/10.1111/j.1365-2117.2006.00310.x
- Meijles, E. W., Kiden, P., Streurman, H. J., van der Plicht, J., Vos, P. C., Gehrels, W. R., & Kopp, R. E. (2018). Holocene relative mean sea -level changes in the Wadden Sea area, northern Netherlands. *Journal of Quaternary Science*, 33(8), 905–923. https://doi.org/10.1002/jqs.3068
- Milker, Y., Nelson, A. R., Horton, B. P., Engelhart, S. E., Bradley, L. A., & Witter, R. C. (2016). Differences in coastal subsidence in southern Oregon (USA) during at least six prehistoric megathrust earthquakes. *Quaternary Science Reviews*, 142, 143–163. https://doi.org/10.1016/j. quascirev.2016.04.017
- Miller, M. M., & Shirzaei, M. (2021). Assessment of future flood hazards for southeastern Texas: Synthesizing subsidence, sea-level rise, and storm surge scenarios. *Geophysical Research Letters*, 48(8), e2021GL092544. https://doi.org/10.1029/2021GL092544
- Minderhoud, P. S. J., Coumou, L., Erkens, G., Middelkoop, H., & Stouthamer, E. (2019). Mekong delta much lower than previously assumed in sea-level rise impact assessments. *Nature Communications*, 10(1), 3847. https://doi.org/10.1038/s41467-019-11602-1
- Minderhoud, P. S. J., Middelkoop, H., Erkens, G., & Stouthamer, E. J. E. R. C. (2020). Groundwater extraction may drown mega-delta: Projections of extraction-induced subsidence and elevation of the Mekong delta for the 21st century. *Environmental Research Communications*, 2(1), 011005. https://doi.org/10.1088/2515-7620/ab5e21
- Nicholls, R. J. (2018). Adapting to sea-level rise. In Z. Zommers & K. Alverson (Eds.), *Resilience: The science of adaptation to climate change* (pp. 13–29). Elsevier.
- Nicholls, R. J., Hanson, S. E., Lowe, J. A., Slangen, A. B. A., Wahl, T., Hinkel, J., & Long, A. J. (2021). Integrating new sea-level scenarios into coastal risk and adaptation assessments: An ongoing process. *Wiley Interdisciplinary Reviews: Climate Change*, 12(3), 1–27. https://doi.org/10. 1002/wcc.706
- Nicholls, R. J., Lincke, D., Hinkel, J., Brown, S., Vafeidis, A. T., Meyssignac, B., et al. (2021). A global analysis of subsidence, relative sea-level change and coastal flood exposure. *Nature Climate Change*, 11(4), 338–342. https://doi.org/10.1038/s41558-021-00993-z
- Ohenhen, L. O., Shirzaei, M., Ojha, C., & Kirwan, M. L. (2023). Hidden vulnerability of US Atlantic coast to sea-level rise due to vertical land motion. *Nature Communications*, 14(1), 2038. https://doi.org/10.1038/s41467-023-37853-7
- Ohenhen, L. O., Shirzaei, M., Ojha, C., Sherpa, S. F., & Nicholls, R. J. (2024). Disappearing cities on US coasts. Nature, 627(8002), 108–115. https://doi.org/10.1038/s41586-024-07038-3
- Park, S.-W., & Hong, S.-H. (2021). Nonlinear modeling of subsidence from a decade of InSAR time series. *Geophysical Research Letters*, 48(3), e2020GL090970. https://doi.org/10.1029/2020GL090970
- Peltier, W. (2004). Global glacial isostasy and the surface of the ice-age Earth: The ICE-5G (VM2) model and GRACE. Annual Review of Earth and Planetary Sciences, 32(1), 111–149. https://doi.org/10.1146/annurev.earth.32.082503.144359
- Peltier, W. R., Argus, D. F., & Drummond, R. (2015). Space geodesy constrains ice age terminal deglaciation: The global ICE-6G_C (VM5a) model. *Journal of Geophysical Research: Solid Earth*, 120(1), 450–487. https://doi.org/10.1002/2014JB011176
- Peltier, W. R., Argus, D. F., & Drummond, R. (2018). Comment on "An assessment of the ICE-6G_C (VM5a) glacial isostatic adjustment model" by Purcell et al. *Journal of Geophysical Research: Solid Earth*, 123(2), 2019–2028. https://doi.org/10.1002/2016jb013844
- Peltier, W. R., & Tushingham, A. M. (1991). Influence of glacial isostatic adjustment on tide gauge measurements of secular sea level change. Journal of Geophysical Research, 96(B4), 6779–6796. https://doi.org/10.1029/90jb02067
- Poitevin, C., Wöppelmann, G., Raucoules, D., Le Cozannet, G., Marcos, M., & Testut, L. (2019). Vertical land motion and relative sea level changes along the coastline of Brest (France) from combined space-borne geodetic methods. *Remote Sensing of Environment*, 222, 275–285. https://doi.org/10.1016/j.rse.2018.12.035

- Raucoules, D., Le Cozannet, G., Woppelmann, G., de Michele, M., Gravelle, M., Daag, A., & Marcos, M. (2013). High nonlinear urban ground motion in Manila (Philippines) from 1993 to 2010 observed by DInSAR: Implications for sea-level measurement. *Remote Sensing of Envi*ronment, 139, 386–397. https://doi.org/10.1016/j.rse.2013.08.021
- Raucoules, D., Parcharidis, I., Feurer, D., Novalli, F., Ferretti, A., Carnec, C., et al. (2008). Ground deformation detection of the greater area of Thessaloniki (Northern Greece) using radar interferometry techniques. *Natural Hazards and Earth System Sciences*, 8(4), 779–788. https://doi. org/10.5194/nhess-8-779-2008
- Sala Calero, J., Vöge, M., Esteves Martins, J., Raucoules, D., de Michelle, M., Vradi, A., & Vecchiotti, F. (2023). EGMS validation report. Retrieved from https://land.copernicus.eu/en/technical-library/validation-report-2015-2021-dataset/@@download/file
- Samsonov, S. V., d'Oreye, N., González, P. J., Tiampo, K. F., Ertolahti, L., & Clague, J. J. (2014). Rapidly accelerating subsidence in the Greater Vancouver region from two decades of ERS-ENVISAT-RADARSAT-2 DInSAR measurements. *Remote Sensing of Environment*, 143, 180– 191. https://doi.org/10.1016/j.rse.2013.12.017
- Schoonbeek, J. B. (1976). Land subsidence as a result of natural gas extraction in the Province of Groningen. SPE European Spring Meeting. https://doi.org/10.2118/5751-MS
- Schuerch, M., Spencer, T., Temmerman, S., Kirwan, M. L., Wolff, C., Lincke, D., et al. (2018). Future response of global coastal wetlands to sealevel rise. *Nature*, 561(7722), 231–234. https://doi.org/10.1038/s41586-018-0476-5
- Shirzaei, M., Freymueller, J., Törnqvist, T. E., Galloway, D. L., Dura, T., & Minderhoud, P. S. J. (2021). Measuring, modelling and projecting coastal land subsidence. *Nature Reviews Earth & Environment*, 2(1), 40–58. https://doi.org/10.1038/s43017-020-00115-x
- Simon, K. M., Riva, R. E. M., Kleinherenbrink, M., & Frederikse, T. (2018). The glacial isostatic adjustment signal at present day in northern Europe and the British Isles estimated from geodetic observations and geophysical models. *Solid Earth*, 9(3), 777–795. https://doi.org/10.5194/ se-9-777-2018
- Slangen, A. B. A., Palmer, M. D., Camargo, C. M. L., Church, J. A., Edwards, T. L., Hermans, T. H. J., et al. (2023). The evolution of 21st century sea-level projections from IPCC AR5 to AR6 and beyond. *Cambridge Prisms: Coastal Futures*, 1(e7), 1–13. https://doi.org/10.1017/cft.2022.8 Spada, G. (2017). Glacial isostatic adjustment and contemporary sea level rise: An overview. *Surveys in Geophysics*, 38(1), 153–185. https://doi. org/10.1007/s10712-016-9379-x
- Stiros, S. C. (2001). Subsidence of the Thessaloniki (Northern Greece) Coastal Plain, 1960-1999. Journal of Engineering Geology, 61(4), 243– 256. https://doi.org/10.1016/s0013-7952(01)00027-8
- Stocchi, P., Spada, G., & Cianetti, S. (2005). Isostatic rebound following the Alpine deglaciation: Impact on the sea level variations and vertical movements in the Mediterranean region. *Geophysical Journal International*, 162(1), 137–147. https://doi.org/10.1111/j.1365-246X.2005. 02653.x
- Teatini, P., Tosi, L., & Strozzi, T. (2011). Quantitative evidence that compaction of Holocene sediments drives the present land subsidence of the Po Delta, Italy. *Journal of Geophysical Research*, 116(B8), B08407. https://doi.org/10.1029/2010jb008122
- Tessler, Z. D., Vörösmarty, C. J., Grossberg, M., Gladkova, I., Aizenman, H., Syvitski, J. P. M., & Foufoula-Georgiou, E. (2015). Profiling risk and sustainability in coastal deltas of the world. *Science*, 349(6248), 638–643. https://doi.org/10.1126/science.aab3574
- Thiéblemont, R. (2024). European CFP EGMS vertical land motion (version 1) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.11091319 Törnqvist, T. E., Wallace, D. J., Storms, J. E. A., Wallinga, J., van Dam, R. L., Blaauw, M., et al. (2008). Mississippi Delta subsidence primarily
- caused by compaction of Holocene strata. *Nature Geoscience*, 1(3), 173–176. https://doi.org/10.1038/ngeo129 Tos, L., Da Lio, C., Strozzi, T., & Teatini, P. (2016). Combining L- and X-Band SAR interferometry to assess ground displacements in het-
- erogeneous coastal environments: The Po River Delta and Venice Lagoon, Italy. *Remote Sensing*, 8(4), 308. https://doi.org/10.3390/rs8040308 Tosi, L., Teatini, P., & Strozzi, T. (2013). Natural versus anthropogenic subsidence of Venice. *Scientific Reports*, 3(1), 2710. https://doi.org/10. 1038/srep02710
- van Asselen, S., Stouthamer, E., & van Asch, T. W. J. (2009). Effects of peat compaction on delta evolution: A review on processes, responses, measuring and modeling. *Earth-Science Reviews*, 92(1–2), 35–51. https://doi.org/10.1016/j.earscirev.2008.11.001
- van Thienen-Visser, K., Sijacic, D., van Wees, J., Kraaijpoel, D., & Roholl, J. (2016). Groningen field 2013 to present Gas production and induced seismicity. TNO.
- Verberne, M., Koster, K., Lourens, A., Gunnink, J., Candela, T., & Fokker, P. A. (2023). Disentangling shallow subsidence sources by data assimilation in a reclaimed urbanized coastal plain, South Flevoland polder, The Netherlands. *Journal of Geophysical Research: Earth Surface*, 128(7), e2022JF007031. https://doi.org/10.1029/2022jf007031
- Vermeersen, B. L. A., Slangen, A. B. A., Gerkema, T., Baart, F., Cohen, K. M., Dangendorf, S., et al. (2018). Sea-level change in the Dutch Wadden Sea. Netherlands Journal of Geosciences, 97(3), 79–127. https://doi.org/10.1017/njg.2018.7
- Wöppelmann, G., Gravelle, M., & Testut, L. (2021). SONEL sea-level observing infrastructure: French contribution to the IUGG Centennial in 2019, and beyond. In C. Boucher (Ed.), *Collection du Bureau des Longitudes* (Vol. 1, pp. 43–53).
- Wöppelmann, G., Le Cozannet, G., de Michele, M., Raucoules, D., Cazenave, A., Garcin, M., et al. (2013). Is land subsidence increasing the exposure to sea level rise in Alexandria, Egypt? *Geophysical Research Letters*, 40(12), 2953–2957. https://doi.org/10.1002/grl.50568
- Wöppelmann, G., & Marcos, M. (2016). Vertical land motion as a key to understanding sea level change and variability. *Review of Geophysics*, 54(1), 64–92. https://doi.org/10.1002/2015RG000502
- World Bank. (2010). Climate risks and adaptation in Asian coastal megacities: A synthesis report.
- Wu, P.-C., Wei, M., & D'Hondt, S. (2022). Subsidence in coastal cities throughout the world observed by InSAR. Geophysical Research Letters, 49(7), e2022GL098477. https://doi.org/10.1029/2022GL098477
- Zanchettin, D., Bruni, S., Raicich, F., Lionello, P., Adloff, F., Androsov, A., et al. (2021). Sea-level rise in Venice: Historic and future trends (review article). *Natural Hazards and Earth System Sciences*, 21(8), 2643–2678. https://doi.org/10.5194/nhess-21-2643-2021
- Zerbini, S., Raicich, F., Prati, C. M., Bruni, S., Del Conte, S., Errico, M., & Santi, E. (2017). Sea-level change in the Northern Mediterranean Sea from long-period tide gauge time series. *Earth-Science Reviews*, *167*, 72–87. https://doi.org/10.1016/j.earscirev.2017.02.009
- Zoccarato, C., Minderhoud, P. S., & Teatini, P. (2018). The role of sedimentation and natural compaction in a prograding delta: Insights from the mega Mekong delta, Vietnam. Scientific Reports, 8(1), 11437. https://doi.org/10.1038/s41598-018-29734-7